Author’s response to reviews

Title: A protocol for a randomised controlled, double-blind feasibility trial investigating the efficacy of fluoxetine treatment in improving memory and learning impairments in patients with mesial temporal lobe epilepsy: Fluoxetine, Learning and Memory in Epilepsy (FLAME trial).

Authors:

Cheney Drew (DrewC5@cardiff.ac.uk)
Mark Postans (PostansM@cardiff.ac.uk)
Catenio Petralia (PetraliaCP@cardiff.ac.uk)
Rachel McNamara (McNamara@cardiff.ac.uk)
Philip Pallmann (PallmannP@cardiff.ac.uk)
Dave Gillespie (GillespieD1@cardiff.ac.uk)
Lisa Evans (EvansLH@cardiff.ac.uk)
Nils Muhlert (Nils.Muhlert@manchester.ac.uk)
Mia Winter (Mia.Winter@wales.nhs.uk)
Khalid Hamandi (HamandiK@cardiff.ac.uk)
William Gray (graywp@cardiff.ac.uk)

Version: 1 Date: 20 Feb 2019

Author’s response to reviews:

Please find attached the above protocol paper, which we have revised in accordance with your initial review.

We have included a brief description of measures of feasibility assessment in the abstract as requested. Further we have included additional detail on the sample size justification on p.6 and on the measures of feasibility assessment that will be used on p.14, again as requested in your e-mail of the 13th March. All changes have been included as tracked changes for ease of review.

We hope the changes we have made are to your satisfaction,
Yours faithfully,

Dr Cheney Drew  
Research Fellow/ Senior Trial Manager  

Prof William Gray  
Chief Investigator